

Citation: Carranza-Rodríguez C, San-Román-Sánchez D, Marrero-Santiago H, Hernández-Cabrera M, Gil-Guillén C, Pisos-Álamo E, et al. (2017) Endomyocardial involvement in asymptomatic sub-Saharan immigrants with helminth-related eosinophilia. PLoS Negl Trop Dis 11(2): e0005403. doi:10.1371/journal. pntd.0005403

Editor: Amy D. Klion, National Institutes of Health, UNITED STATES

Received: September 29, 2016

Accepted: February 12, 2017

Published: February 24, 2017

Copyright: © 2017 Carranza-Rodríguez et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper.

Funding: The authors received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Endomyocardial involvement in asymptomatic sub-Saharan immigrants with helminth-related eosinophilia

Cristina Carranza-Rodríguez^{1,2®}*, Daniel San-Román-Sánchez^{1,3®}, Héctor Marrero-Santiago³, Michele Hernández-Cabrera^{1,2}, Carlos Gil-Guillén³, Elena Pisos-Álamo^{1,2}, Nieves Jaén-Sánchez^{1,2}, José-Luis Pérez-Arellano^{1,2}

1 Department of Medical and Surgery Sciences, University of Las Palmas de Gran Canaria, Las Palmas de Gran Canaria, Spain, 2 Infectious Diseases and Tropical Medicine Division, Complejo Hospitalario Universitario Insular Materno Infantil, Las Palmas de Gran Canaria, Spain, 3 Cardiology Division, Complejo Hospitalario Universitario Insular Materno Infantil, Las Palmas de Gran Canaria, Spain, 3 Cardiology Division, Complejo Hospitalario Universitario Insular Materno Infantil, Las Palmas de Gran Canaria, Spain

So These authors contributed equally to this work.

* cristinacarranzarodriguez@gmail.com

Abstract

Background

Among immigrants of sub-Saharan origin, parasitic infection is the leading cause of eosinophilia, which is generally interpreted as a defense mechanism. A side effect of the inflammatory mediators released by eosinophils is damage to host organs, especially the heart. The main objectives of this study were to i) assess cardiac involvement in asymptomatic sub-Saharan immigrants with eosinophilia, ii) relate the presence of lesions with the degree of eosinophilia, and iii) study the relationship between cardiac involvement and the type of causative parasite.

Methodology/Principle findings

In total, the study included 50 black immigrants (37 patients and 13 controls) from sub-Saharan Africa. In all subjects, heart structure and function were evaluated in a blinded manner using Sonos 5500 echocardiographic equipment. The findings were classified and described according to established criteria. The diagnostic criteria for helminthosis were those reported in the literature. Serum eosinophil-derived neurotoxin levels were measured using enzyme-linked immunosorbent assay. A significant association was found between the presence of eosinophilia and structural alterations (mitral valve thickening). However, the lack of an association between the degree of eosinophilia and heart valve disease and the absence of valve involvement in some patients with eosinophilia suggest the role of other factors in the appearance of endocardial lesions. There was also no association between the type of helminth and valve involvement.

Conclusions

We, therefore, suggest that transthoracic echocardiography be performed in every sub-Saharan individual with eosinophilia in order to rule out early heart valve lesions.

Author summary

Endomyocardial fibrosis is characterized by fibrosis of the apical endocardium of the right ventricle, left ventricle, or both. Epidemiological studies of endomyocardial fibrosis indicate a predominance in tropical regions, with young people and men being affected predominantly. Little is known about the natural history and pathogenic factors of this condition. One of the most important factors is the presence of eosinophilia, which is mainly related to helminth infections. The aim of the study was to evaluate cardiac involvement in patients with absolute eosinophilia. We performed an echocardiographic study in asymptomatic sub-Saharan immigrants with eosinophilia and compared them with a group of control patients from the same region and of the same age. Our results suggest that eosinophilia associated with helminth infection (regardless of the causative microorganism) contributes to the asymptomatic phases of endomyocardial involvement. Treatment of these infections is simple, effective, and economical. Therefore, we suggest that all immigrant patients with eosinophilia undergo etiologic study and causal treatment as well as transthoracic echocardiography in order to rule out early damage.

Introduction

Eosinophilia is a common finding among immigrants, especially those of sub-Saharan origin. In this population, the main cause of eosinophilia is parasitic infection, specifically by helminths [1]. There now are sufficient data that demonstrate that eosinophils, through their proinflammatory mediators such as eosinophil-derived neurotoxin (EDN) and eosinophil cationic protein, play an important role in the control of the larval phase of these parasitic diseases [2,3]. Thus, helminth-induced eosinophilia may be seen as a defense mechanism.

A side effect of the inflammatory mediators released by eosinophils is lesions in host organs, with cardiac involvement being especially characteristic [4,5]. Injury to the heart structures associated with eosinophilia may take two clinical forms: acute and subacute or chronic. The acute forms follow two main patterns: acute necrotizing eosinophilic myocarditis and hypersensitivity myocarditis [5,6]. The subacute or chronic forms also have two polar forms: Davies' endomyocardial fibrosis, found in tropical areas, and Loeffler's fibroblastic endocarditis, which occurs in temperate regions [4,7]. Regardless of the clinical form, cardiac involvement in parasitic-induced eosinophilia has been documented in epidemiological studies as well as in experimental clinical and animal models [8–10].

Therefore, the objectives of this study were to i) evaluate cardiac involvement in asymptomatic sub-Saharan immigrants with eosinophilia; ii) relate the presence of lesions with the degree of eosinophilia; and iii) study the relationship between cardiac involvement and the type of causative parasite.

Methods

Ethics statement

This study was approved by the Ethics Committee of the Complejo Hospitalario Insular-Materno Infantil (Las Palmas de Gran Canaria, Spain). Written informed consent was obtained from the participants in their local languages. The reasons for the collection and storage of blood samples and echocardiography were clearly described in the consent form.

Table 1. Demographic Data and General Characteristics.

	Study Group (Eosinophils >450 cells/µL)	Control Group (Eosinophils <450 cells/µL)	p-Value
Patients (n)	37	13	-
Age, years (Mean; SD)	25 (6)	26 (3)	0.54*
Sex (Male/Female)	34/3	13/0	0.56**
Country (Alphabetical order)			
Cameroon	5	1	
Ghana	7	2	
Guinea	2	0	
Guinea-Bissau	1	1	
Equatorial Guinea	2	0	
Mali	5	4	
Mauritania	0	1	
Nigeria	2	0	
The Gambia	1	0	
Sierra Leone	12	3	
Тодо	0	1	0.36***
Eosinophilia (Mean; SD)	932.1 ± 443.8	218.3 ± 108.9	-
Mild (450–999/µL),	26 (70.2%)	-	
Moderate (1000–1499/µL)	7 (19%)	-	
Severe (>1500/µL)	4 (10.8%)	-	
Parasitic infection			
Schistosomal species	8 (21.6%)	-	
Species causing filariasis	13 (35.1%)	-	
Geohelminths	7 (18.9%)	-	
Co-parasitized	4 (10.8%)	-	

*Student's t-test

**Fisher's exact test

***Chi-square test

doi:10.1371/journal.pntd.0005403.t001

Study population

In total, 52 black immigrants from sub-Saharan Africa, namely West Africa, were included in the study. Their mean age was 25 years and the male/female ratio was 47/5. Two patients were excluded, owing to rheumatic mitral valve disease in one case and mitral valve prolapse associated with mild mitral valve insufficiency in the other. Therefore, the final study group consisted of 37 patients and 13 controls.

The subjects, who met following inclusion criteria, were recruited consecutively during external consultations at the Unit of Infectious Diseases and Tropical Medicine: i) absence of clinical manifestations of heart disease, ii) presence of absolute eosinophilia (>450/ μ L), and iii) explicit consent to participate in the study. Eosinophilia was classified as mild (450–999/ μ L), moderate (1,000–1,499/ μ L), or severe (>1,500/ μ L).

The control group consisted of asymptomatic subjects from the same geographic region in whom the presence of eosinophilia and/or parasitosis were ruled out by means of the indicated tests, who provided consent to participate in the study. <u>Table 1</u> summarizes the demographic characteristics of each group.

Parasitological techniques

All parasitological studies (direct and serological) were performed in the same specialized laboratory using the investigation strategy suggested by Whetham for African patients, with some modifications [11].

The direct parasitological tests used in this study were i) three examinations of stool samples for ova and parasites using the Kato-Katz and Ritchie techniques and specific tests for *Strongyloides stercoralis* (Baermann concentration test and agar culture method) [12], ii) microscopy of a terminal urine sample, and iii) Knott's test for the detection of microfilaremia. The Immune Chromatographic Test (ICT Filariasis Binax, Portland, Maine) was used for the detection of *Wuchereria bancrofti* antigens, with skin snips and the Mazzotti test used in selected cases.

The serological assays were performed using enzyme-linked immunosorbent assay (ELISA). Crude extracts of adult *Dirofilaria immitis* worm (AWA Di) [13], *Schistosoma bovis* adult worm antigens (AWA Sb) [14], *F. hepatica* excretory/secretory antigens (E/S Fh) [15], and *Trichinella spiralis* L1 worm antigens (L1 Ts) [16] were used, respectively, for the detection of filariasis, schistosomiasis, fascioliasis, and trichinellosis. Polystyrene microtiter plates (Costar) were coated with 100 μ L of antigens per well at a previously determined protein concentration in carbonate buffer (pH 9.6). Serum at a dilution of 1:100 was added to the wells and incubated for 1 h at 37°C. Horseradish peroxidase goat anti-human immunoglobulin G (Sigma, St. Louis, MO, USA) was added at different dilutions. Washes were performed three times with 200 μ L of phosphate buffered saline-Tween 20 per well. After incubation for 1 h at 37°C, the substrate solution (*ortho*-phenylenediamine plus H₂O₂) was added and the reaction was stopped with 3N H₂SO₄. The individual conditions of the assays and the sensitivity and specificity of the tests are described in our previous publication [1].

Echocardiographic studies

In order to assess the degree and type of cardiac involvement in the study and control group subjects, two-dimensional, M-mode, and Doppler evaluations were performed using the same echocardiographic equipment (Sonos 5500; Philips Medical Systems, Andover, MA, USA). The tracings were recorded for playback later on and blinded analysis by a single investigator.

The following determinations were made for each patient: i) The dimensions of the left and right ventricular chambers and the left atrium were measured using the M-mode technique in the parasternal long and short axis planes. ii) The morphological characteristics of the heart were analyzed using two-dimensional echocardiography (2D echo), with optimization of the image by adjusting the frequency and parameters of gain, compression, and image processing. Standard parasternal long and short axes as well as four- and two-chamber apical and subcostal planes were used. A high-resolution zoom magnification system was applied to specific areas, including the ventricular apical zones and valvular planes, for better visualization and quantification of possible alterations. iii) The left ventricular ejection fraction was analyzed using 2D echo, using the Simpson biplane technique. iv) The functional parameters of ventricular filling were analyzed using pulsed-wave Doppler ultrasound of transmitral and transtricuspid flow during mid-expiratory apnea, with calculation of the initial peak velocities (E in cm/s), slope deceleration (TD in milliseconds), speed atrial contraction (A in cm/s), and E/A ratio. If possible, the pulmonary venous flow (apical four-chamber plane) with calculation of the peak systolic (S in cm/s) and diastolic (D in cm/s) velocities and S/D ratio, and peak velocity (cm/s) and duration (in milliseconds) of the retrograde atrial wave (Ar) were measured. In all cases, measurements of a minimum of four consecutive heartbeats were analyzed. v) The presence of valve regurgitation was evaluated using color-coded Doppler ultrasound in the

parasternal planes, long and short axes, and apical four and two chambers. vi) In cases with tricuspid insufficiency, the pulmonary systolic pressure was calculated using the Bernoulli equation: $4V^2+10$ (in mm Hg) (where V represents the maximum velocity).

The findings were classified into two groups according to the following descriptions from the literature on cardiac involvement in the presence of eosinophilia [17,18]:

i) **Classic findings:** Endocardial thickening, described as thickening of the wall, with refringent and additional echogenic layers in the endocardial zone; thrombus in the left ventricle, characterized by an echodense mass at the left cardiac apex, causing obliteration or limitation of the blood flow during filling; apical thrombus in the right ventricle, with the same characteristics as on the left side; affectation (thickening) of the mitral valve leaflets, especially the posterior leaflet, determined by thickness values greater than 3 mm; and tricuspid insufficiency, defined as the detection of signals of tricuspid valve insufficiency on color-coded Doppler with a semiquantitative assessment of its severity

ii) **Non-classic findings:** Left ventricular hypertrophy, defined according to the criteria described by Casale et al. [19] using measurements of interventricular septum thickness (IST) and posterior wall thickness (PWT) (the degree of left ventricular hypertrophy was classified as mild [IST = 12-13 mm; PWT = 11-12 mm], moderate [IST = 13-14 mm; PWT = 12-13 mm], or severe [IST = 14-15 mm; PWT = 13-14 mm]); left ventricular dilatation, defined as ventricular transverse diastolic diameter >56 mm; pericardial effusion, the presence of echo-free spaces of more than 5 mm in the left ventricular posterior wall; hyperdynamic left ventricle, characterized by a ventricular ejection fraction equal to or greater than 70%; and severe left ventricular diastolic dysfunction, defined as an E/A ratio of mitral transvalvular flow >2 and/ or TD <130 milliseconds.

Determination of serum EDN levels

Serum EDN levels were measured by sandwich ELISA using a commercial kit (EDN ELISA kit ref 7630, MBL International Corporation, USA) following the manufacturer instructions. The limit of detection of the technique is 0.62 ng/mL [20]. The normal values are \leq 66.6 ng/mL [21].

Statistical methods

Qualitative variables were expressed as frequency distributions. Quantitative variables with normal distributions were expressed as the mean and standard deviation (SD). The associations between qualitative variables were analyzed using the chi-square or Fisher's exact test, as appropriate. Normally distributed variables were compared using the Student's t-test. Data distribution as compared with theoretical models was verified, and the assumption of the homogeneity of variance was tested in all cases. Analyses were performed using GraphPad InStat. Differences were considered significant if p values were less than 0.05.

Results

In the study group, the degree of eosinophilia was mild in 26 (70%), moderate in seven (22%), and severe in four (8%) patients. The etiological diagnosis was filariasis in 13 cases, schistosomiasis in eight cases, coexistence of filariasis and schistosomiasis in four cases, a geohelmintic disease in seven cases, and indeterminable in five people. The relationship between the degree of eosinophilia and the causal diagnosis is shown in Table 2.

All the subjects had an adequate echocardiographic image and were in sinus rhythm, with an average heart rate of 69 beats/minute (SD: 11). All subjects had normal blood pressure, with a mean value of 125/75 mm Hg. Tables $\underline{3}$ and $\underline{4}$ show the qualitative and quantitative

	n	Eosinophil Counts (Mean; SD)	Final Diagnosis
Mild (450–999/µL)	26	721 (444)	Filarial diseases (9) Schistosomal diseases (8) Geohelmintic diseases (5) Undetected (4)
Moderate (1,000–1,499/ μL)	7	1,119 (63)	Filarial diseases (2) Filarial/Schistosomal diseases (2) Geohelmintic diseases (2) Undetected (1)
Severe (>1,500/µL)	4	1,975 (515)	Filarial diseases (2) Filarial/Schistosomal diseases (2)

Table 2. Types of Eosinophilia and Final Diagnoses.

doi:10.1371/journal.pntd.0005403.t002

echocardiographic results as well as their statistical analyses. As can be observed, a significant association was found only between thickening of the mitral valve and the presence of eosino-philia, with the thickness of the mitral valve being significantly greater in cases with eosino-philia than in those without eosinophilia. Two representative images are shown in Fig 1.

The serum concentration of EDN was determined in 29 patients with eosinophilia, 28 of whom had high values (>66.6 ng/mL). However, there were no significant differences (p = 0.46) in the serum concentration of this protein between patients with valve involvement (115.5 ng/mL, SD: 36.6) and those without it (129.1 ng/mL, SD: 53.8). On the other hand, there was no statistically significant association between the number of eosinophils and serum levels of EDN (p = 0.18).

Further, there was no statistical association between the degree of eosinophilia and the presence of valve involvement, or between the type of helminth responsible for eosinophilia and valve involvement, as shown in Table 5.

Discussion

Endomyocardial fibrosis (EMF) is a syndrome characterized by fibrosis of the apical endocardium of the right ventricle, left ventricle, or both. The clinical manifestations are mainly related to the restriction of ventricular filling, which may trigger cardiac failure.

Epidemiological studies of EMF indicate a predominance in tropical regions (15° around the equator), with young people (with a bimodal peak at 10 and 30 years of age) and men

	Study Group (Eosinophils >450 cells/µL)	Control Group (Eosinophils <450 cells/µL)	p-Value
Classic findings			
Endocardial thickening	2/37	0/13	1.0*
Left ventricle apical thrombus	1/37	0/13	1.0*
Right ventricle apical thrombus	0/37	0/13	1,0*
Posterior mitral leaflet thickening	18/37	0/13	0.001*
Tricuspid valve thickening	7/37	0/13	0.16*
Tricuspid insufficiency	15/37	4/13	0.08**
Aortic valve thickening	2/37	0/13	1.0*
Non-classic findings			
Left ventricular hypertrophy	3/37	0/13	1.0*
Dilated left ventricle	0/37	0/13	1.0*
Hyperdynamic left ventricle	1/37	0/13	1.0*
Pericardial effusion	0/37	0/13	1.0*
Severe left ventricular diastolic dysfunction	0/37	0/13	1.0*

Table 3. Qualitative Echocardiographic Findings.

* Fisher's exact test

** Chi-square test

doi:10.1371/journal.pntd.0005403.t003

	Study Group (Eosinophils >450 cells/µL)	Control Group (Eosinophils <450 cells/µL)	p-Value
Posterior mitral leaflet thickening	2.98 (0.67) mm	2.47 (0.39) mm	0.002
Tricuspid valve thickening	2.37 (0.69) mm	2.04 (0.49) mm	0.079
Aortic valve thickening	1.89 (0.56) mm	1.75 (0.31) mm	0.32
Left ventricular ejection fraction	56.67 (4.91) %	56.38 (3.93) %	0.83
DTI (E/A) of the mitral annulus	2.02 (0.55)	2.12 (0.73)	0.64
Pulmonary venous flow (S/D)	1.13 (0.31)	1.11 (0.38)	0.86
Mitral valve E/A ratio	1.72 (0.59)	1.76 (0.52)	0.81
DTI (E/A) of the basal septum	1.57 (0.38)	1.46 (0.32)	0.38
Tricuspid valve E/A ratio	1.92 (0.48)	1.80 (0.43)	0.45
DTI (E/A) of the middle lateral wall	1.84 (0.46)	1.90 (0.28)	0.59
DTI (E/A) of the middle septum	1.64 (0.43)	1.60 (0.31)	0.77
IVRT	74.61 (13.63) ms	74.07 (11.95) ms	0.89
Color M-mode flow propagation velocity	56.32 (18.06) cm/s	50.11(12.12) cm/s	0.19

Table 4	Quantitative	Echocardiog	raphic Fir	ndings (Mea	n and Standar	d Deviation)
10010 11		Longoaraiog		laningo (inioa	in anna otannaan	a bottaaton/

DTI, Doppler tissue imaging; IVRT, isovolumetric relaxation time.

doi:10.1371/journal.pntd.0005403.t004

being affected predominantly. In this sense, our series is similar to that of patients with EMF in terms of age and sex, although our series included a greater number of people from West Africa. In some series, EMF data were available for up to 19.8% of the population studied by echocardiography, and only 22.7% of these had symptoms [22]. These data suggest that EMF clinical series mainly include patients in advanced stages of disease.

The pathogenesis of EMF is not fully understood, although several types of factors have been identified. One of the most important factors is the presence of eosinophilia related to helminth infections or other causes (e.g., hypereosinophilic syndrome, Churg-Strauss disease, reactions to drugs, or neoplasia) [23–25]. The association between the presence of eosinophilia and EMF varies greatly depending on the geographical region (e.g., highest in Uganda [26], low in India [27]). The other factors involved are nutritional (low protein diet, cassava consumption) [7], environmental (increased cerium) [27], genetic [22], and immunological [28].



Fig 1. Characteristic Echocardiographic Findings. (A) Two-dimensional echocardiography image: parasternal long axis showing a thickening of the aortic sigmoid (arrows) without altering its function. (B) Parasternal long axis view of the left ventricle, showing slight thickening of the mitral leaflets (arrows), the most frequent finding in this study.

doi:10.1371/journal.pntd.0005403.g001

Valve Involvement Schistosoma spp.		Species causing filariasis	Geohelminth	Total
Yes	8	6	1	15
No	3	9	6	18
Total	11	15	7	33

Table 5. The Association Between Valve Involvement and the Type of Helminth.

Fisher's exact test (Freeman-Halton extension), p = 0.07

doi:10.1371/journal.pntd.0005403.t005

Therefore, the first objective of our study was to compare the echocardiographic findings of asymptomatic sub-Saharan immigrants from West Africa with eosinophilia with a group of controls from the same region and of equivalent age. Our results indicate that there is a significant association between the presence of eosinophilia and structural alterations, in the absence of evident clinical manifestations. Thus, mitral valve thickening suggests early involvement, independent of the racial characteristics and other factors controlled for in the study (e.g., age or sex). There were significant differences in the structure of the leaflets of the mitral valve, with a significantly greater thickness in cases with eosinophilia. These data may have clinical relevance since the predominant involvement of the mitral valve in patients with EMF and eosinophilia has been demonstrated in other studies as well [29–33]. On the other hand, it is interesting to note that eosinophilia and/or the presence of parasitosis in this context do not give rise to alterations in left ventricular contractile function (expressed as a percentage of the ejection fraction) or alter the global or regional diastolic function of the left or right side of the heart.

Although eosinophilia plays an important role in the pathogenesis of EMF, there is no direct relationship between the two. Thus, in our series, we observed on the one hand that a significant percentage of patients with eosinophilia did not present any data suggesting cardiac involvement; on the other hand, there was no association between the degree of eosinophilia and cardiac valve involvement. One possible explanation for these findings is that not only is an increase in eosinophils required for the development of EMF but that these are also activated and release inflammatory mediators [34]. However, our data do not support this interpretation since in nearly all patients with eosinophilia, there was an increase in serum EDN levels. In summary, eosinophilia plays an important role in the development of EMF, although other factors (e.g., genetics, nutritional factors, duration of helminthic infection) can contribute to its appearance [35].

In our series, the cause of eosinophilia was identified by direct or indirect techniques in 86% of cases. In all cases, it was a helminthic disease, mainly filariasis, schistosomiasis, or both. Cardiac involvement in helminths has three main patterns: i) myocarditis due to direct or indirect myocardial involvement in *Trichinella* spp. infection [36]; ii) acute endocarditis in *Toxocara* spp. infection [37,38], and iii) endomyocardial fibrosis. Infection with several helminths is associated with EMF, although infection by different species of *Filarias* is the most frequently described association [39–41]. However, in our series, there was no statistical association between valve involvement and the type of parasite. In fact, we observed this association with *Schistosoma* spp. infection, which is rarely indicated in the literature [42–44].

These data suggest that eosinophilia related to helminth infection (regardless of the causative microorganism) contributes to the asymptomatic phases of endomyocardial lesions. The treatment of these infections is simple, effective, and cheap. Therefore, we suggest that an etiological study and causal treatment, as well as transthoracic echocardiography, be performed in all immigrant patients with eosinophilia in order to rule out early lesions.

Author Contributions

Conceptualization: JLPA CCR.

Data curation: CCR.

Formal analysis: DSRS.

Investigation: CCR.

Methodology: DSRS JLPA CCR.

Project administration: CCR JLPA.

Resources: MHC EPA NJS HMS CGG JLPA.

Supervision: JLPA DSRS.

Validation: DSRS JLPA CCR.

Visualization: CCR JLPA.

Writing - original draft: CCR JLPA.

Writing - review & editing: CCR JLPA.

References

- Pardo J, Carranza C, Muro A, Angel-Moreno A, Martín AM, Martín T, et al. Helminth-related eosinophilia in African immigrants, Gran Canaria. Emerg Infect Dis. 2006; 12: 1587–1589. doi: <u>10.3201/eid1210</u>. 060102 PMID: <u>17176579</u>
- 2. Behm CA, Ovington KS. The role of eosinophils in parasitic helminth infections: insights from genetically modified mice. Parasitol Today. 2000; 16: 202–209. PMID: 10782080
- Meusen ENT, Balic A. Do eosinophils have a role in the killing of helminth parasites? Parasitol Today. 2000; 16: 95–101. PMID: 10689327
- 4. Touze JE, Fourcade L, Heno P, Mafart B, Mourot S. The heart and the eosinophil. Med Trop (Mars). 1998; 58 (Suppl 4): 459–464.
- Parrillo JE. Heart disease and the eosinophil. N Engl J Med. 1990; 323: 1560–1561. doi: 10.1056/ NEJM199011293232211 PMID: 2233938
- Grabellus F, Mall G, Schnabel PA, Wieneke H, Pfeifer U, Kersting C, et al. Immunohistochemical differentiation of eosinophilic heart diseases using antibodies against eosinophil activation markers. Histopathology. 2005; 46: 89–97. doi: 10.1111/j.1365-2559.2005.01999.x PMID: 15656891
- Grimaldi A, Mocumbi AO, Freers J, Lachaud M, Mirabel M, Ferreira B, et al. Tropical endomyocardial fibrosis: natural history, challenges, and perspectives. Circulation. 2016; 133: 2503–2515. doi: 10.1161/ CIRCULATIONAHA.115.021178 PMID: 27297343
- Rutakingirwa M, Ziegler JL, Newton R, Freers J. Poverty and eosinophilia are risk factors for endomyocardial fibrosis (EMF) in Uganda. Trop Med Int Health. 1999; 4: 229–235. PMID: 10223220
- Andy JJ, Ogunowo PO, Akpan NA, Odigwe CO, Ekanem IA, Esin RA. Helminth associated hypereosinophilia and tropical endomyocardial fibrosis (EMF) in Nigeria. Acta Trop. 1998; 69: 127–140. PMID: 9588233
- Arima M, Kanoh T. Eosinophilic myocarditis associated with dense deposits of eosinophil cationic protein (ECP) in endomyocardium with high serum ECP. Heart 1999; 81: 669–671. PMID: 10336931
- Whetham J, Day JN, Armstrong M, Chiodini PL, Whitty CJ. Investigation of tropical eosinophilia; assessing a strategy based on geographical area. J Infect. 2003; 46: 180–185. PMID: 12643868
- Siddiqui AA, Berk SL. Diagnosis of *Strongyloides stercoralis* infection. Clin Infect Dis. 2001; 33: 1040– 1047. doi: 10.1086/322707 PMID: 11528578
- Perera L, Pérez Arellano JL, Cordero M, Simón F, Muro A. Utility of antibodies against a 22 kDa molecule of *Dirofilaria immitis* in the diagnosis of human pulmonary dirofilariosis. Trop Med Intern Health. 1998; 3: 151–155.
- 14. Pardo J, Carranza C, Turrientes MC, Pérez Arellano JL, Lopez-Velez R, Ramajo V, et al. Utility of *Schistosoma bovis* adult worm antigens for diagnosis of human schistosomiasis by enzyme-linked

immunosorbent assay and electroimmuno-transfer blot techniques. Clin Diagn Lab Immunol. 2004; 11: 1165–1170. doi: 10.1128/CDLI.11.6.1165-1170.2004 PMID: 15539523

- Hillyer GV, Soler de Galanes M. Identification of a 17 kDa Fasciola hepatica immunodiagnostic antigen by enzyme-linked immunoelectrotransfer blot technique. J Clin Microbiol. 1988; 26: 2048–2053. PMID: 3182993
- Alcántara P, Correa D. Human humoral immune response against *Trichinella spiralis*. Int J Parasitol. 1993; 23: 657–660. PMID: 8225769
- Acquatella H, Schiller NB, Puigbo JJ, Gomez-Mancebo JR, Suarez C, Acquatella G. Value of twodimensional echocardiography in endomyocardial disease with and without eosinophilia. A clinical and pathologic study. Circulation. 1993; 67: 1219–1226.
- Ommen SR, Seward JB, Tajik AJ. Clinical and echocardiographic features of hypereosinophilic syndromes. Am J Cardiol. 2000; 80: 110–113.
- Casale PN, Devereux RB, Milner M, Zullo G, Harshfield GA, Pickering TG, et al. Value of echocardiographic measurement of left ventricular mass in predicting cardiovascular morbid events in hypertensive men. Ann Intern Med. 1986; 105: 173–178. PMID: 2942070
- Morioka J, Kurosawa M, Inamura H, Nakagami R, Mizushima Y, Chihara J, et al. Development of a novel enzyme-linked immunosorbent assay for blood and urinary eosinophil-derived neurotoxin: a preliminary study in patients with bronchial asthma. Int Arch Allergy Immunol. 2000; 122: 49–57. PMID: 10859469
- Liu GT, Hwang CS, Hsieh CH, Lu CH, Chang SL, Lee JC, et al. Eosinophil-derived neurotoxin is elevated in patients with amyotrophic lateral sclerosis. Mediators Inflamm. 2013; 2013: 421389. doi: <u>10.1155/2013/421389</u> PMID: 23533305
- Mocumbi AO, Ferreira MB, Sidi D, Yacoub MH. A population study of endomyocardial fibrosis in a rural area of Mozambique. N Engl J Med. 2008; 359: 43–49. doi: <u>10.1056/NEJMoa0708629</u> PMID: 18596273
- 23. Spry CJ, Take M, Tai PC. Eosinophilic disorders affecting the myocardium and endocardium: a review. Heart Vessels Suppl. 1985; 1: 240–242. PMID: 3916474
- 24. Pinar-Sopena J, Candell-Riera J, San-José-Laporte A, Bosch Gil J, García del Castillo H, Vilardell Tarres M, et al. Manifestaciones ecocardiográficas en pacientes con hipereosinofilia. Rev Esp Cardiol. 1990; 43: 450–456.
- Yalonetsky S, Mishaly D, Ben-Barak A, Lorber A. Mitral valve destruction by Hodgkin's lymphoma-associated Loefler endocarditis. Pediatr Cardiol. 2008; 29: 983–985. doi: <u>10.1007/s00246-007-9135-6</u> PMID: 18004615
- Freers J, Masembe V, Schmauz R, Mayanja-Kizza H. Endomyocardial fibrosis syndrome in Uganda. Lancet. 2000; 355: 1994–1995. doi: 10.1016/S0140-6736(05)72932-9 PMID: 10859061
- Valiathan SM, Kartha CC. Endomyocardial fibrosis-the possible connexion with myocardial levels of magnesium and cerium. Int J Cardiol. 1990; 28: 1–5. PMID: 2194985
- Mocumbi AO, Latif N, Yacoub MH. Presence of circulating anti-myosin antibodies in endomyocardial fibrosis. PLoS Negl Trop Dis. 2010; 4: e661. doi: 10.1371/journal.pntd.0000661 PMID: 20422043
- Pavlovic M, Berdat P, Holzer B, Aebi C, Carrel T, Pfammatter JP. Severe mitral valve involvement in a child with hypereosinophilia secondary to parasitic infection. J Heart Valve Dis. 2003; 12: 649–651. PMID: 14565720
- Jánoskuti L, Förhécz Z, Lengyel M. Unique presentation of hypereosinophilic syndrome: recurrent mitral prosthetic valve thrombosis without endomyocardial disease. J Heart Valve Dis. 2006; 15: 726–729. PMID: 17044383
- Madhwal S, Goldberg J, Barcena J, Guha A, Gogate P, Cmolik B, et al. Unusual cause of acute mitral regurgitation: idiopathic hypereosinophilic syndrome. Ann Thorac Surg. 2012; 93: 974–977. doi: <u>10</u>. 1016/j.athoracsur.2011.08.062 PMID: 22364989
- Fang Q, Gu TX, Xiu ZY, Zhang ZW. Bioprosthetic valve replacement in idiopathic hypereosinophilic syndrome with mitral valve involvement. J Thorac Cardiovasc Surg. 2013; 146: e45–47. doi: <u>10.1016/j.</u> jtcvs.2013.07.033 PMID: 24001347
- Zakhama L, Slama I, Boussabah E, Harbegue B, Mimouni M, Abdelaali N, et al. Recurrent native and prosthetic mitral valve thrombosis in idiopathic hypereosinophilic syndrome. J Heart Valve Dis. 2014; 23: 168–170. PMID: 25076546
- Tai PC, Ackerman SJ, Spry CJ, Dunnette S, Olsen EG, Gleich GJ. Deposits of eosinophil granule proteins in cardiac tissues of patients with eosinophilic endomyocardial disease. Lancet. 1987; 1: 643–647. PMID: 2882081
- Patel AK, D'Arbela PG, Somers K. Endomyocardial fibrosis and eosinophilia. Br Heart J. 1977; 39: 238–241. PMID: 849383

- Puljiz I, Beus A, Kuzman I, Seiwerth S. Electrocardiographic changes and myocarditis in trichinellosis: a retrospective study of 154 patients. Ann Trop Med Parasitol. 2005; 99: 403–411. doi: 10.1179/ 136485905X36307 PMID: 15949188
- Abe K, Shimokawa H, Kubota T, Nawa Y, Takeshita A. Myocarditis associated with visceral larva migrans due to Toxocara canis. Intern Med. 2002; 41: 706–708. PMID: 12322796
- Enko K, Tada T, Ohgo KO, Nagase S, Nakamura K, Ohta K, et al. Fulminant eosinophilic myocarditis associated with visceral larva migrans caused by *Toxocara canis infection*. Circ J. 2009; 73: 1344– 1348. PMID: 19122304
- Berenguer A, Plancha E, Muñoz Gil J. Right ventricular endomyocardial fibrosis and microfilarial infection. Int J Cardiol. 2003; 87: 287–289. PMID: 12559553
- Nutman TB, Miller KD, Mulligan M, Ottesen EA. Loa loa infection in temporary residents of endemic regions: recognition of a hyperresponsive syndrome with characteristic clinical manifestations. J Infect Dis. 1986; 154: 10–18. PMID: 3458832
- Fux CA, Chappuis B, Holzer B, Aebi C, Bordmann G, Marti H, et al. *Mansonella perstans* causing symptomatic hypereosinophilia in a missionary family. Travel Med Infect Dis. 2006; 4: 275–280. doi: 10.1016/j.tmaid.2005.07.001 PMID: 16905458
- 42. Bustinduy AL, Luzinda K, Mpoya S, Gothard P, Stone N, Wright S, et al. Endomyocardial fibrosis (EMF) in a Ugandan child with advanced hepatosplenic schistosomiasis: coincidence or connection? Am J Trop Med Hyg. 2014; 91: 798–780. doi: 10.4269/ajtmh.14-0156 PMID: 25002295
- 43. Carneiro Rde C, Santos AL, Brant LC, Rabelo FT, Ligeiro CM, Barcelos IP, et al. Endomyocardial fibrosis associated with mansoni schistosomiasis. Rev Soc Bras Med Trop. 2011; 44: 644–645. PMID: 22031086
- 44. Gran F, Albert DC, Moreno A. Schistosomiasis and tropical endomyocardial fibrosis with pulmonary hypertension. Rev Esp Cardiol. 2011; 64: 713. doi: 10.1016/j.recesp.2010.02.003 PMID: 21592640